

CPS 13th - 20th Sep

- Friday:
 - TOF requested a low intensity beam (50-100e10p)
 - RP access of over 1h in East Irrad (Charm not aware)
- Week-end:
 - Problem in booster extraction, 1/2h beam down
 - POPS tripped, 1/2h beam down
- Technical Stop:
 - Beams off on Monday at 16h
 - Some devices still locked out at scheduled restart time
 - Problems restarting septa 57&61 and F61N.QFO02: PIPO and First line intervention
 - BSF54 stuck in the beam line
 - Beam back for EAST @18h but unstable spill
 - Power OFF/ON this morning of F61.QFO03 solves the regulation issue